Author's response to reviews

Title: Effectiveness of the Dader Method for Pharmaceutical Care in patients with Bipolar I Disorder. EMDADER-TAB: study protocol for a randomized controlled trial

Authors:

Andrea Salazar-Ospina (asalazarospina@gmail.com)
Pedro Amariles (pamaris@farmacia.udea.edu.co)
Dora M Benjumea-Gutierrez (doraben@farmacia.udea.edu.co)
Francisco J Gutierrez-Henao (fgutier@une.net.co)
Luis F Rodríguez (dirmed@clinicasanjuandedios.com)
Maria J Faus (mfaus@ugr.es)

Version: 4 Date: 1 April 2014

Author's response to reviews: see over
Medellin, April 1, 2014.

Drs.
Doug Altman
Curt Furberg
Jeremy Grimshaw
Editors-in-Chief
TRIALS JOURNAL

Subject: Response to reviewers' comments of MS: 3967372931004697 - Effectiveness of Dader Method for Pharmaceutical Care in patients with Bipolar I Disorder. EMDADER-TAB: Randomized Controlled Trial - NCT01750255

Dear editors.

By logging on to the system, we have resubmitted the re-reviewed article “Effectiveness of Dader Method for Pharmaceutical Care in patients with Bipolar I Disorder. EMDADER-TAB: Randomized Controlled Trial - NCT01750255”. The manuscript has been edited according to comments of the Senior Editor. Thus, All changes (including deletions) to the text had been underlined or highlighted, and a addressing each numbered comment (with page numbers indicating the location of change).

Editorial request: Please ensure the title conforms to journal style for study protocol articles. The title should follow the format ?___________: study protocol for a randomized controlled trial.?

Title has been rewrite as [Effectiveness of the Dader Method for Pharmaceutical Care in patients with Bipolar I Disorder. EMDADER-TAB: study protocol for a randomized controlled trial]

COMMENT(S)
1. P(age) 4, p(aragraph) 1, l(ine) 11. Suggest inserting [have] between [to] and [poor].
   [have] has been inserted between [to] and [poor]. P(age) 4, p(aragraph) 1, l(ine) 11.

2. P 6, p 1, l 6. Delete one [.].
   One [.] has been deleted. 2. P 6, p 1, l 6.
3. P 14, p 2, l 4. Delete [them] and 4. P 14, p 2, l 5. Suggest rewrite as […] the study, from months …].

[The period of recruitment of the patients is 18 months, and those patients included during this time, will be them evaluated. Once enrollment period ends, we have 12 months to end with recent reviews of patients to the study. From months 13 to 30 while the study is continuing. In final 12 months of study following outcomes will not be assessed.] Has been rewrite as [The recruitment period will be of 18 months, and patients will be evaluated during 12 months, starting from the date of their recruitment. Once the enrolment period ends, final 12 months will be only used to evaluate the latest patients] P 14, p 2, l 2-5.

5. P 15, p 2, l 1. Delete []).

[]) has been deleted. P 15, p 2, l 1.


[Multivariate] has been replacing by [Multivariable]. P 17, p 1, l 22.


[and/] has been deleted. P 18, p 2, l 6

8. P 22, p 1, l 4. There is a need to show that [†] denotes [deceased]. It is not currently in the text.

[Deceased] has been denoted in author affiliations. P 1, after L corresponding author

9. P 22, R(ereference) 1, l 4. Delete [[] and []], ie, the notation should be (Spanish).

[ ] has been deleted. P 22, p 1, l 4

10. P 23, R 8, Insert [[] before the first word of the title which appears to be [National] and insert [] after the last word of the title which appears to be [2003]. Insert the location of the Ministry with a city and country.

[ ] has been inserted between [National] and [2003]. Has been included the location of the Ministry [Bogota]. P 23, R8.

11. P 25, R 19, l 4. Rewrite as [Columbian].

[Columbian] has been rewrite as [Columbian]. P 25, R 19, l 4.

12. P 25, R 20, l 1. Replace [et al] by more authors up to 30 before using [et al].

[et al] has been deleted and was rewrite as [Lopez MC, Ospina-Duque J, Ruiz-Linares A.] P 25, R 20, l 1.
[2] has been deleted and was rewrite as [2nd Ed.] P 26, R 28, l 1.

14. P 28, R 41, l 2. Rewrite as [studies.]].
[Pharmaceutical Care: Concepts, processes and [studies] Has been rewrite as [Pharmaceutical Care: Concepts, processes and studies].

I appreciate so much your collaboration and contribution to improve our article.

Prof. Pedro Amariles
University of Antioquia